Author's response to reviews

Title: Intraarticular Hemorrhage Due To Bevacizumab In A Patient With Metastatic Colorectal Cancer

Authors:

Mukremin Uysal (mukreminuysal@yahoo.com)
Sema S Goksu (drsemagoksu@yahoo.com.tr)
Hasan S Coskun (hscoskun@yahoo.com)
Burhan Savas (bsavas@akdeniz.edu.tr)
Mustafa Ozdogan (ozdoganmd@yahoo.com)
Hakan Bozcuk (hbozcuk@gmail.com)

Version: 2 Date: 3 November 2011

Author's response to reviews:

Please include a Consent section at the end of the manuscript, before the reference list. It is OK

- Please include the study design in your title, i.e. Case report. For example: A presenting with B in C: a case report. It is OK

- Please include patient’s ethnicity in the abstract and case presentation sections. It is OK

- Please replace the header case report to “Case Presentation.”. It is OK

- Please include a competing interests section at the end of the manuscript, before the reference list. It is OK

- If applicable, please include an acknowledgement section at the end of the manuscript before the reference list. It is OK

- Please include the authors contribution section. It is OK

- Please edit your figure to preserve the anonymity of the patient by removing or blacking out the writings found. It is OK